Author's response to reviews

Title: Adult-onset leukoencephalopathy, brain calcifications and cysts: a case report

Authors:

- Gang Cheng (yjscg2003@126.com)
- Yaming Wang (yamingwang@sohu.com)
- Chao Dong (flyingperson85@126.com)
- Jianning Zhang (jnzhang2005@gmail.com)
- Yuhong Meng (584056134@qq.com)

Version: 3 Date: 26 March 2013

Author's response to reviews: see over
November 15th, 2012

Neurosurgery Department
Tele: +86 010 66958404
Fax: +86 010 66958404

Dear Editor:

We are enclosing a manuscript entitled “A case of adult-onset leukoencephalopathy with calcifications and cysts” to be considered for publication in Journal of Medical Case Reports. Leukoencephalopathy, brain calcifications, and cysts (LCC) is a rare syndrome. This is an example of adult onset of LCC and expanded the clinical spectrum of features of LCC. The content of this manuscript, in part or in full, has not been published elsewhere in any form. It has been reviewed and approved by all the authors. I certify that this manuscript is a unique submission and is not being considered for publication with any other source in any medium. I am responsible for the authenticity of the manuscript and take full responsibility for the data, the analyses and interpretation, and the conduct of the research. We have no declarations of financial interest. We have no conflict of interest. All the data in the manuscript adhere to ethical standards in health and medical care. Please forward all comments to me at the address shown below. If there are shortcomings about the manuscript, please inform us as soon as possible. Thank you for your consideration and we look forward to a favorable review.
Reviewer's report and answers
- in Wargon et al, the patient was not symptom free but presented a lacunar stroke revealed by hemiplegia.
Answer: the case report has been deleted in the manuscript
- some reports of LCC are lacking, including:
Answer: in the instructions for authors, there should be no more than 15 references. Some reports of LCC are not included in the manuscript.

Please also address the following formatting changes required for your paper:
1. Please include the ethnicity of the patient in the case presentation section of the manuscript.
Answer: the ethnicity of the patients has been included in the manuscript.
2. Please replace the term “man” with “male”
Answer: “man” has been replaced with “male” in the manuscript.
3. Please remove any text from figure one which may reveal the patient’s identity (e.g. patient’s number, patient’s name, dates, etc)
Answer: the text that may reveal the patient’s identity has been removed from figure.
4. Please remove the specific dates found in the Case presentation section
Answer: the specific dates found in the Case presentation section has been removed.

Sincerely,

Jianning Zhang, MD, PhD
Chairman and professor
Department of Neurosurgery
PLA Navy General Hospital
6# Fuchenglu Road, Beijing 100048, P.R. China

Tele: +86 010 66958404
Fax: +86 010 66958404
Email: jnzhang2005@gmail.com